Author’s response to reviews

Title: Parathyroidectomy improves cardiovascular risk factors in normocalcemic and hypercalcemic primary hyperparathyroidism

Authors:

Selvihan Beysel (beyselselvihan@gmail.com)
Mustafa Caliskan (mcaliskan37@yahoo.com)
Muhammed Kizilgul (muhammedkzgl@gmail.com)
Mahmut Apaydin (drmahmut@yandex.com)
Seyfullah Kan (seyfullah76@hotmail.com)
Mustafa Ozbek (mustafaozbek10@hotmail.com)
Erman Cakal (ermancakal@hotmail.com)

Version: 2 Date: 23 Apr 2019

Author’s response to reviews:

Dear Sir

Special thanks for accepting our manuscript and improving our paper.

English editing was made by native English man.

We made revision according to reviewers.

Thanks